Author's response to reviews

Title: Durability of Antiretroviral Therapy and Predictors of Virologic Failure Among Perinatally HIV-Infected Children in Tanzania: A Four-Year Follow-Up

Authors:

Dorothy E Dow (dorothy.dow@duke.edu)
Aisa M Shayo (aisa.shayo@gmail.com)
Coleen K Cunningham (coleen.cunningham@duke.edu)
Elizabeth A Reddy (elizabeth.reddy@duke.edu)

Version: 2 Date: 8 April 2014

Author's response to reviews:

Thank you for noticing these oversights. Please now find:

1) line numbering and page numbering added

2) title page, abstract, added to the manuscript document. Figure captions moved above tables.

3) key words deleted from the abstract provided separate to the manuscript